Author's response to reviews

Title: Gastro-enteritis outbreak among Nordic patients with psoriasis in a health centre in Gran Canaria, Spain: a cohort study

Authors:

Hanne M Eriksen (hanne.merete.eriksen@fhi.no)
Philippe J Guerin (Philippe.GUERIN@epicentre.msf.org)
Karin Nygard (kany@fhi.no)
Marika Hjertqvist (Marika_Hjertqvist/SMI/SE%SMI@smi.ki.se)
Birgitta de Jong (birgitta.de.jong@smi.ki.se)
Angela M.C Rose (a.rose@invs.sante.fr)
Markku Kuusi (markku.kuusi@ktl.fi)
Ulrike Durr (ulistep@club-internet.fr)
Amos G Rojas (agarroj@gobiernodecanarias.org)
Cato Mork (cato.mork@klinmed.uio.no)
Preben Aavitsland (Preben.Aavitsland@fhi.no)

Version: 5 Date: 28 October 2004

Author's response to reviews: see over
Dear Sir

I have now done the formatting changes required.

1. The number next to each author has been formatted in superscript.
2. In the article is it stated that 'The authors declare that they have no competing interests'.
3. The tables are formatted to fit in the page in portrait format.
4. The figure title is part of the manuscript after the reference list.
5. Figures are cropped as closely as possible to minimise white space around the image.

I hope I have understood the requested format changes correct this time..

Kindly yours

Hanne Eriksen